Author’s response to reviews

Title: Individual and healthcare system factors influencing antenatal care attendance in Saudi Arabia

Authors:

Wdad Alanazy (806810@swansea.ac.uk)

Amy Brown (a.e.brown@swansea.ac.uk)

Version: 2 Date: 19 Nov 2019

Author’s response to reviews:

Thank you for your helpful and constructive comments for the feedback below. We have made all changes, ensuring in particular accuracy in the table. We have edited down the text we added in last time in response to reviewer one and have clarified all queries.

In response to reviewer two queries below, the revised text includes means and standard deviations and percentages for the significant results. We could add percentages and means to the tables, but this would make the tables much larger, increasing space, for non-significant results. We are happy to do this if required.

Table 3: Please also show n (%) for every bivariate analysis of chi-squared test, not only the chi-squared value.

Table 4: Please show mean (SD) for every bivariate analysis of t test, not only the t-value.